%**start of header
\def\title{Homework 3, February 5}
\input common.tex
%**end of header

\bs
These problems are due Wednesday, February 14.

Do p.~365, problems 1, (2), 3, (5), (6), 7, 8, (9), 10, 13; and p.~369,
problems 4, (5), (6), (7), 9, 10.

The exercises in parentheses are to be turned in only if you are uncertain
whether you got the correct answer.

\bye


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Here we set the local variables for my emacs text editor.
%   Local Variables:
%   TeX-command: "amstex"
%   TeX-start-of-header: "%**start of header"
%   TeX-end-of-header:   "%**end of header"
%   End:
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
